Author’s response to reviews

Title: Renal capsule metastasis from renal pelvic cancer: a case report

Authors:

Yasuyuki Kobayashi (ya_su_koba@yahoo.co.jp)
Hiroki Arai (arai_h@kich.itami.hyogo.jp)
Masahito Honda (honda_m@kich.itami.hyogo.jp)
Takashi Matsumoto (matsumoto_t@kich.itami.hyogo.jp)
Kyotaro Yoshida (yoshida_k@kich.itami.hyogo.jp)

Version: 3 Date: 20 Feb 2018

Author’s response to reviews:

20 February 2018

Vitaly Margulis, MD
Editorial Board, BMC Urology

Dear Tillie Cryer:

Thank you for your recent e-mail and the provisional acceptance of our manuscript BURO-D-17-00182R1 entitled “Renal capsule metastasis from renal pelvic cancer: a case report”. Please find enclosed our revised manuscript for reconsideration of publication as a case report in BMC Urology.

We appreciate your comments and make some revisions.

1. Form of consent

Written informed consent was obtained from the patient for publication of this manuscript.

2. Authors’ contributions - TM further justification needed

TM made substantial contributions to analysis and interpretation of data.
3. Remove CARE checklist and cover letter additional files

We remove the CARE checklist and cover letter.

4. Clean copy/remove red highlighting

I upload final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Thank you for reconsidering our manuscript for publication in BMC Urology.

Sincerely,

Yasuyuki Kobayashi, MD

The Department of Urology
Kinki Central Hospital of Mutual Aid Association of Public Teachers
3-1 Kurumazuka, Itami, Hyogo 664-8533, Japan
Phone: +81-72-781-3712; Fax: +81-72-779-1567
E-mail: ya_su_koba@yahoo.co.jp